Author’s response to reviews

Title: Microbicides Development Programme: Engaging the community in the standard of care debate in a vaginal microbicide trial in Mwanza, Tanzania

Authors:

Andrew J Vallely (a.vallely@uq.edu.au)
Charles Shagi (Charles.Shagi@Amref.org)
Shelley Lees (shelley.lees@lshtm.ac.uk)
Katherine Shapiro (shapiro.kathy@gmail.com)
Joseph Masanja (josephmasanja@yahoo.com)
Lawi Nikolau (josephmasanja@yahoo.com)
Johari Kazimoto (josephmasanja@yahoo.com)
Selephina Soteli (nyandashagi@gmail.com)
Claire Moffat (claire.moffat@lshtm.ac.uk)
John Changalucha (jchangalucha@yahoo.com)
Sheena McCormack (smc@ctu.mrc.ac.uk)
Richard J Hayes (richard.hayes@lshtm.ac.uk)

Version: 5 Date: 6 October 2009

Author’s response to reviews:

A minor error has been corrected in the Methods section (final paragraph, Study Population sub-section, page 5):

'A total of 1146 HIV sero-negative subjects have been enrolled into the main trial in Mwanza, with follow-up expected to be completed by mid 2009'

has been changed to:

'A total of 1146 HIV sero-negative subjects have been enrolled into the main trial in Mwanza, with follow-up completed in mid-2009'